Author’s response to reviews

Title: The use of Matrigel has no influence on tumor development or PET imaging in FaDu human head and neck cancer xenografts

Authors:
Frederikke Petrine Fliedner (frederikkepf@sund.ku.dk)
Anders Hansen (aeha@sund.ku.dk)
Jesper Jørgensen (jtran@sund.ku.dk)
Andreas Kjær (akjaer@sund.ku.dk)

Version: 1 Date: 11 Nov 2015

Author’s response to reviews:
See attached Cover Letter for our response to comments.

Two revised manuscripts are attached; one revised manuscript were changes are tracked, and one final version.

The manuscript with changes tracked are uploaded as supplementary for the reviewers.